

Beaumont, H; Augood, C; Duckitt, K; Lethaby, A (2002) Danazol for heavy menstrual bleeding. The Cochrane database of systematic reviews, 2 (2). CD001017. ISSN 1469-493X DOI: https://doi.org/10.1002/14651858.CD001017

Downloaded from: http://researchonline.lshtm.ac.uk/11167/

DOI: 10.1002/14651858.CD001017

Usage Guidelines

Please refer to usage guidelines at http://researchonline.lshtm.ac.uk/policies.html or alterna $tively\ contact\ research on line@lshtm.ac.uk.$

Available under license: Copyright the author(s)

Danazol for heavy menstrual bleeding (Review)

Beaumont H, Augood C, Duckitt K, Lethaby A



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2002, Issue 2

http://www.thecochranelibrary.com



TABLE OF CONTENTS

EADER	1
BSTRACT	1
AIN LANGUAGE SUMMARY	2
ACKGROUND	2
BJECTIVES	3
ESULTS	3
ISCUSSION	7
JTHORS' CONCLUSIONS	9
CKNOWLEDGEMENTS	0
EFERENCES	0
DURCES OF SUPPORT	1
IDEX TERMS	2

[Intervention Review]

Danazol for heavy menstrual bleeding

H Beaumont, C Augood, K Duckitt, A Lethaby

Contact address: Dr Heather Beaumont, medical student, 3 Hazelhurst Road, Llandafff, Cardiff, Wales, CF14 2FW, UK. beaumont@cardiff.ac.uk.

Editorial group: Cochrane Menstrual Disorders and Subfertility Group. **Publication status and date:** Unchanged, published in Issue 2, 2007. **Review content assessed as up-to-date:** .

Citation: Beaumont H, Augood C, Duckitt K, Lethaby A. Danazol for heavy menstrual bleeding. *Cochrane Database of Systematic Reviews* 2002, Issue 2. Art. No.: CD001017. DOI: 10.1002/14651858.CD001017.

Copyright © 2007 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Heavy menstrual bleeding (HMB) is an important cause of ill health in pre menopausal women. Medical therapy, with the avoidance of possibly unnecessary surgery is an attractive treatment option, but there is considerable variation in practice and uncertainty about the most effective therapy. Danazol is a synthetic steroid with anti-oestrogenic and anti progestogenic activity, and weak androgenic properties. Danazol suppresses oestrogen and progesterone receptors in the endometrium, leading to endometrial atrophy (thinning of the lining of the uterus) and reduced menstrual loss and to amenorrhoea in some women.

Objectives

To determine the effectiveness and tolerability of danazol when used for heavy menstrual bleeding in women of reproductive years.

Search strategy

We searched the Menstrual Disorders and Subfertility Group's Specialised Register of controlled trials (6 Nov 2001). We also searched the Cochrane Controlled Trials Register (Cochrane Library, Issue 3, 2001), MEDLINE (1966 to Oct 2001), EMBASE (1980 to Oct 2001), Current Contents (1993 to Oct 2001), CINAHL (1982 to Sept 2001), and the National Research Register (Issue 3, 2001). Attempts were also made to identify trials from citation lists of included trials and relevant review articles. In most cases the first author of each included trial was contacted for unpublished additional information.

Selection criteria

Randomised controlled trials of danazol versus placebo, any other medical (non-surgical) therapy or danazol in different dosages for heavy menstrual bleeding in women of reproductive age with regular HMB measured either subjectively or objectively. Trials that included women with post menopausal bleeding, intermenstrual bleeding and pathological causes of heavy menstrual bleeding were excluded.

Data collection and analysis

Nine RCTs, with 353 women, were identified that fulfilled the inclusion criteria for this review. Quality assessment and data extraction were performed independently by two reviewers. The main outcomes were menstrual blood loss, the number of women experiencing adverse effects, weight gain, withdrawals due to adverse effects and dysmenorrhoea. If data could not be extracted in a form suitable for meta-analysis, they were presented in a descriptive format.

Main results

Most data were not in a form suitable for meta analysis, and the results are based on a small number of trials, all of which are underpowered. Danazol appears to be more effective than placebo, progestogens, NSAIDs and the OCP at reducing MBL, but confidence intervals were wide. Treatment with danazol caused more adverse events than NSAIDs (OR 7.0; 95% CI 1.7, 28.2) and progestogens (OR 4.05, 95% CI 1.6, 10.2), but this did not appear to affect adherence to treatment. Danazol was shown to significantly lower the duration of menses when compared with NSAIDs (WMD -1.0; 95% CI -1.8, -0.3) and a progesterone releasing IUD (WMD -6.0; 95% CI -7.3, -4.8). There were no randomised trials comparing danazol with tranexamic acid or the levonorgestrel-releasing intrauterine system.

Authors' conclusions

Danazol appears to be an effective treatment for heavy menstrual bleeding compared to other medical treatments, though it is uncertain whether it is acceptable to women. The use of danazol may be limited by its side effect profile, its acceptability to women and the need for continuing treatment. Overall no strong recommendations can be made due to the small number of trials, and the small sample sizes of the included trials.

PLAIN LANGUAGE SUMMARY

Danazol is an effective treatment for the reduction of heavy menstrual bleeding, but the adverse effects may be unacceptable to women

Options to help avoid surgery can be important for many women who are having problems with heavy menstrual bleeding. One of the drug options is danazol. Danazol suppresses the hormones that increase the endometrium (the lining of the uterus that is shed during menstruation). However, danazol can also produce male characteristics and some menopause-like symptoms, as well as weight gain and acne. The review found that although danazol is effective at reducing menstrual blood loss there are not enough trials to show whether this treatment is acceptable to women with heavy menstrual bleeding.

BACKGROUND

Heavy menstrual bleeding (menorrhagia) can have a significant impact on women's lives. In the UK, one in 20 women aged 30-49 consult their general practitioner each year with heavy menstrual bleeding (Vessey 1992) and it accounts for 12% of all gynaecological referrals (Bradlow 1992). Once referred to a gynaecologist, surgical intervention is highly likely (Coulter 1991).

Heavy menstrual bleeding (HMB) or menorrhagia is clinically defined as greater than or equal to 80 ml blood loss per menstrual cycle (Hallberg 1966; Cole 1971). It is, however, the woman's perception of her own menstrual loss which is the key determinant in a referral and, indeed, subsequent treatment. The main aim of the treatment of menorrhagia is to reduce blood loss in order to improve quality of life and prevent anaemia.

Many factors can cause HMB for example coagulation disorders, endocrine disorders, uterine abnormalities and other pelvic pathology. These disorders should be excluded before decisions are made about treatment as they may require different management. However, in most cases, there is no pathological cause of the heavy

bleeding, and the condition is labelled dysfunctional uterine bleeding (EHCB 1995). Eighty percent of women treated for menorrhagia have no uterine abnormality and over a third of the women undergoing hysterectomies for HMB have a normal uterus removed (Gath 1982; Clarke 1995). Although patient satisfaction with hysterectomies is high (Coulter 1994), there are complications and occasional death associated with hysterectomy (Dicker 1982). Complications are more likely when the hysterectomy is performed by the open abdominal route, as is usually the case (Hospital 1995). Effective medical therapy, that avoids unnecessary surgery, is therefore an attractive alternative.

A wide variety of medications are available to reduce HMB but their effectiveness has been questioned (Coulter 1995). The aim of this review is to see if danazol is an effective therapy for HMB. Danazol is chemically derived from testosterone (a naturally occurring hormone). It inhibits ovulation and reduces oestrogen levels. It also causes endometrial atrophy (thinning of the lining of the uterus), reduced menstrual loss and leads to amenorrhoea (absence of periods) in some women (Chimbira 1980b). Danazol has

a dramatic effect on increasing haemoglobin and serum ferritin levels and may therefore be valuable in women who need effective therapy to stop very heavy bleeding and restore their haemoglobin and iron status to normal (Ford 1994; Chimbira 1979). Danazol has androgenic properties (a tendency to cause male characteristics) which may result in acne, seborrhoea (greasy skin) and hirsutism (excessive hair growth). Other side effects include weight gain, irritability, musculoskeletal pains, hot flushes and breast atrophy (loss of breast tissue). Longer term treatment with danazol may cause liver effects (including benign hepatic adenomas) in some women.

OBJECTIVES

To determine the effectiveness and tolerability of danazol when given for heavy menstrual bleeding in women of reproductive years.

We wished to investigate:

- 1. Whether treatment with danazol is more effective than placebo in reducing heavy menstrual blood loss.
- 2. Whether treatment with danazol is more effective than other medical therapies (antifibrinolytics, NSAIDs, progestogens) in reducing heavy menstrual blood loss.
- 3. If effective, what is the optimum dosage of danazol.
- 4. Whether treatment with danazol leads to an improved quality of life for women with heavy menstrual blood loss.
- 5. Whether women tolerate treatment with danazol and find it an acceptable treatment.

RESULTS

Overall nine studies compared the use of danazol with placebo, other medical treatments or different doses of danazol for the treatment of heavy menstrual bleeding. The studies contained a total of 353 participants.

DANAZOL VS. PLACEBO

One study with 66 participants compared danazol 200 mg once daily with a matched placebo once daily for three months of treatment (Lamb 1987).

Menstrual blood loss (objective/subjective)

This study did not assess objective MBL, and subjective MBL was assessed using an unidentified scoring system. It was not possible to include data for the menstrual blood loss scores, as the type of scoring system used and the figures were poorly reported in the paper. The authors report no significant differences in blood loss scores for the placebo group when comparing before and after

treatment scores. A significant difference in blood loss scores was reported for the danazol group compared to the pre-treatment scores, but it is unclear how this was calculated.

Duration of menses

The data for duration of menses could not be included in this review as the figures were inadequately reported in the paper. The author reported no difference in duration of menses for the placebo group comparing pre and post treatment figures. A significant difference in duration of menses was reported for the danazol group (when comparing before and after treatment figures), but the authors do not indicate how this was calculated.

Withdrawals due to side effects

The number of withdrawals due to side effects during the intervention did not significantly differ between the two groups (OR 2.06, 95% CI 0.18, 23.94).

Body weight

The trial reported data on the mean body weight for each group. After three months treatment the mean weight (kg) of the danazol group was significantly greater than that of the placebo group (WMD 6.70, 95% CI 0.98, 12.42).

DANAZOL VS. PROGESTOGENS

Five of the included studies involving a total of 131 women compared 200 mg danazol with a progestogen. For four of the studies the comparison intervention was norethisterone, and in one study the progestogen was medroxyprogesterone acetate.

Menstrual blood loss (objective/subjective)

Four of the studies comparing danazol with a progestogen reported data on MBL. In two trials MBL was measured objectively using the alkaline haematin method. One small parallel trial comparing danazol with norethindrone in the meta analysis (n=37) showed no significant difference between the two groups for menstrual blood loss after treatment (WMD -35.60, 95% CI -102.20, 31.00) (Higham 1993). The other trial reporting objectively measured MBL, contained data which was not reported in a form suitable for pooling. This trial reported MBL as medians and ranges, rather than means and standard deviations and is included as descriptive data in the Other Data section (Cameron 1987). In this trial the groups were not comparable at baseline, and the study compared MBL after treatment to MBL at baseline for the different treatment groups. There was no significant difference between the before and after treatment values for the progestogen group (p>0.05), where as MBL after treatment was significantly lower than that at baseline for the danazol group (p<0.05) (Cameron 1987).

Two studies comparing danazol with a progestogen reported subjective measures of MBL. Both are included as descriptive data in the other data section due to skewing of the data in one (Dunphy 1998) and use of a non-standard bleeding scale in the

other (Bonduelle 1991). One trial used the pictorial chart method described by Higham (1990) to record monthly blood loss and reported that MBL after three months treatment was significantly lower in the danazol group compared to the medroxyprogesterone acetate group (p=0.0128) (Dunphy 1998). The other study used a seven point scoring system where daily bleeding scores were combined to give a score for each menstrual period. The trial compared bleeding scores after three months treatment and showed a significant difference between the two groups in favour of danazol (p<0.05).

One study (n=18) included the outcome of MBL three months after the intervention. The trial showed that MBL (assessed by the pictorial chart method) was significantly lower in the progestogen group three months after treatment was stopped (WMD 203.00, 95%CI 25.65, 380.35) (Dunphy 1998).

Side effects

Four trials reported the number of women in each of the two treatment groups experiencing side effects. The four studies in the meta analysis showed that significantly more women in the danazol group experienced side effects compared to the progestogen group (OR 4.05, 95% CI 1.61, 10.21) (Bonduelle 1991; Buyru 1995; Dunphy 1998; Higham 1993). Commonly reported side effects for danazol treatment included acne, weight gain, headache, nausea and tiredness. One study found that adverse effects were reported with a similar frequency and were of a similar nature in both treatment groups (Bonduelle 1991). Commonly reported side effects in this study were weight gain, bloating, gastro-intestinal symptoms, skin changes, lethargy, depression and reduced concentration. Buyru 1995 found that headaches and muscle cramps were reported with similar frequency in both groups, but the danazol group also complained of weight gain, acne, nausea and intermenstrual bleeding. The Higham (1993) study found that both groups reported the adverse effects of muscle cramps and depression with similar frequency. In this study, other commonly reported side effects amongst the danazol group included headache, weight gain, nausea and vomiting and acne, where as other adverse effects reported amongst the progestogen group were premenstrual tension symptoms (Higham 1993).

Withdrawals due to side effects

Four trials comparing danazol with a progestogen reported the number of withdrawals due to side effects. Pooling of data from these studies showed that there was no significant difference between the two groups in terms of withdrawals due to side effects (OR 1.67, 95% CI 0.53, 5.23).

Duration of menses

Four studies comparing danazol with a progestogen assessed the duration of menses. The Chi square test for heterogeneity showed there is significant heterogeneity within the comparison (19.52, df=3, p=0.0002). To consider this heterogeneity, the data was anal-

ysed using a random effects model to take into account the variability between the studies when calculating the summary statistic. Pooling of data from these trials shows that there was no significant difference between the groups in the duration of menses after treatment (WMD -0.74, 95%CI -2.31, 0.82).

Dysmenorrhoea

One of the studies comparing danazol with a progestogen reported this outcome (Bonduelle 1991). Abdominal pain and backache were assessed using a three point scoring system. The study compared before and after treatment scores for both measures of pain for the two groups. For the danazol group there was no significant difference in the before and after treatment scores on either of these measures of dysmenorrhoea (p>0.05). The post treatment backache score was significantly lower than that at baseline for the norethisterone group (p<0.05), but there was no significant difference in the before and after treatment abdominal pain scores (p>0.05). This trial was included as descriptive data in the other data section due to the use of a non standard pain scale.

Weight gain

Three studies comparing danazol with a progestogen reported weight gain as an outcome measure. One trial reported weight gain as mean weight gain, and showed that the mean weight gain was significantly higher in the danazol group (WMD 2.80, 95% CI 1.60, 4.00) (Dunphy 1998). One study reported the number of women with a weight gain of >2kg and the another reported the number of women with weight gain of >3kg. For weight gain as a dichotomous variable there was no significant difference between the danazol and progestogen groups. Where the number of women with weight gain >2kg was assessed the OR was 2.86 (95% CI 0.48, 17.11) (Higham 1993) and where the numbers with weight gain >3kg was reported the OR was 5.57 (95% CI 0.48, 64.09) (Bonduelle 1991). But the reported results are imprecise with wide confidence intervals.

Efficacy of intervention

Three studies included this outcome. One study assessed this objectively, reporting the number in each group with MBL of <80 ml at the end of the intervention, and the other two studies assessed efficacy subjectively. The study assessing efficacy objectively showed a significant difference in favour of danazol. Significantly more women in the danazol group had a MBL of less than 80 ml at the end of the intervention (OR 7.20 95% CI 1.28, 40.37) (Higham 1993).

For subjective efficacy of medication, where efficacy was measured as the number of women rating the treatment as highly or moderately effective, subjective efficacy was significantly better after danazol treatment than after norethisterone (OR 4.33, 95% CI 1.09, 17.17) (Higham 1993). Where efficacy was assessed by the numbers rating their MBL as none or moderate, there was no significantly described by the numbers rating their MBL as none or moderate, there was no significantly described by the numbers rating their MBL as none or moderate, there was no significantly described by the numbers rating their MBL as none or moderate, there was no significantly described by the numbers rating their MBL as none or moderate, there was no significantly described by the numbers rating the numbers of the numbers of

nificant difference between the two groups (OR 5.83, 95% CI 0.70, 48.87) (Bonduelle 1991).

DANAZOL VS. NSAIDS

Three studies compared danazol with a NSAID for the treatment of heavy menstrual bleeding. All three studies compared danazol with mefenamic acid, but one trial also used naproxen as a comparison intervention.

Menstrual blood loss (objective/subjective)

All three studies reported this outcome, and all assessed MBL using the alkaline haematin method. One small parallel trial (n=39) in the meta-analysis showed that mean MBL after two months treatment was significantly lower in the danazol group compared to the mefenamic acid group (WMD -96.70, 95% CI -138.80, -54.60) (Dockeray 1989). Two other trials are included as descriptive data in the other data section as one reported MBL in a form unsuitable for inclusion in the meta analysis and in the other the data showed significant skewness. One trial included in the other data section, compared danazol with mefenamic acid and reported MBL as medians and ranges. In this trial the groups were not comparable at baseline and the study compared MBL after treatment to MBL at baseline for the different treatment groups, rather than comparing MBL across the groups. MBL after treatment was significantly lower than that at baseline for the danazol group (p<0.05), but there was no significant difference between the before and after treatment MBL values for the mefenamic acid group (p>0.05) (Cameron 1987). The other trial was a crossover trial which compared danazol with both mefenamic acid and naproxen. When the mean and standard deviation were calculated from the individual participant data and put into the meta-analysis, mean MBL after two months treatment was significantly lower in the danazol group compared to both the mefenamic acid group (p=0.001) and the naproxen group (p=0.02) (Fraser 1991). These figures refer to data prior to participants crossing over to the other treatment.

Side effects

One study which compared danazol with mefenamic acid reported this outcome. There were significantly more adverse effects in the danazol group compared to the mefenamic acid group 75% compared to 30% (OR 7.00, 95% CI 1.74, 28.17) (Dockeray 1989). The mefenamic acid group complained mainly of nausea, vomiting and diarrhoea; the danazol group complained of more serious adverse effects such as musculoskeletal pains, dizziness, flushes, acne, behavioural changes, tiredness, breast atrophy, hirsutism and hoarseness.

Withdrawals due to side effects

None of the studies comparing danazol with a NSAID reported this outcome.

Duration of menses

Two trials in both of which mefenamic acid was the comparison intervention assessed this outcome. Pooling of data from these trials showed the duration of menses was significantly shorter in the danazol groups after two months of treatment (WMD -1.03; 95% CI -1.78 -0.28).

Dysmenorrhoea

One parallel study reported this outcome. Dysmenorrhoea was assessed in two ways in the study; according to the number of women in each intervention group who reported an improvement in dysmenorrhoea after two months treatment, and using a three point pain scale scoring system. The number reporting an improvement in dysmenorrhoea showed no significant difference between the two interventions (OR 1.20; 95% CI 0.20, 7.31). The dysmenorrhoea scores showed no significant difference between the two groups after two months of treatment (p>0.05) (Dockeray 1989). The data for the dysmenorrhoea scores was included in the Other Data section due to the use of a non-standard pain scale.

Acceptability of treatment

One study included the outcome of the numbers in each group unwilling to continue that particular treatment. This small parallel trial (n=39) compared danazol with mefenamic acid and showed no significant difference between the two groups with regard to the number unwilling to continue the treatment (OR 1.11; 95% CI 0.32, 3.90) (Dockeray 1989).

DANAZOL VS. ORAL CONTRACEPTIVE

One small cross-over study involving 12 women compared danazol 200 mg with an oral contraceptive (ethinyl oestradiol 30ug and levonorgestrel 150ug) (Fraser 1991). Only the data prior to the cross-over were included in the analysis.

Menstrual blood loss (objective/subjective)

Blood loss was measured objectively using the alkaline haematin method. After two months treatment mean MBL for the danazol group was significantly lower than that for the oral contraceptive group (p=0.02). The trial was included in the Other Data section as descriptive data due to skewness of the data.

Other outcomes

The only trial comparing danazol with an oral contraceptive did not report any other outcome measures.

DANAZOL VS. PROGESTERONE RELEASING IUD

One trial with 14 participants compared danazol with a progesterone releasing intrauterine device (releasing 65ug progesterone daily) for two months treatment.

Menstrual blood loss (objective/subjective)

MBL was measured objectively using the alkaline haematin method. The study compared MBL after the intervention with

that at baseline for each of the interventions, rather than making comparisons between the groups as the groups were not comparable at baseline. For both groups MBL after the intervention was significantly lower than that at baseline p<0.05 and p<0.01 for the danazol and progesterone releasing IUD interventions respectively (Cameron 1987). The trial was included as descriptive data in the Other Data section as the data was not reported in a form suitable for inclusion in the meta-analysis.

Duration of menses

The one trial comparing danazol with a progesterone releasing IUD reported data on the duration of menses after two months treatment compared with that at baseline for each group. The results show that the mean duration of menses was significantly shorter in the danazol group (WMD -6.00; 95% CI -7.25, -4.75).

DANAZOL 200 mg VS. DANAZOL 100 mg

One study involving 32 participants compared danazol 200 mg with danazol 100 mg for the treatment of heavy menstrual bleeding (Chimbira 1980a).

Menstrual blood loss (objective/subjective)

MBL was reported objectively after three months treatment and compared with mean MBL at baseline for each group. For both groups mean MBL after three months treatment was significantly lower than mean MBL at baseline (p<0.01). There was no significant difference between the two groups after three months of treatment (p=0.2) (Chimbira 1980a). This trial was reported in the Other Data section as descriptive data as the data showed significant skewness.

Duration of menses

The mean duration of menses was included as one of the outcome measures in the study comparing danazol 200 mg with danazol 100 mg. The authors reported that the duration of menses after three months treatment with danazol 200 mg was significantly less than that at baseline (p<0.01). There was no significant difference in the duration of menses after three months treatment with danazol 100 mg compared to that at baseline (p>0.05). The trial was reported as descriptive data in the Other Data section as the figures were inadequately reported in the paper.

Dysmenorrhoea

The study assessed the outcome of the numbers in each group reporting an improvement in dysmenorrhoea after three months treatment. For this outcome there was no significant difference between the two groups (OR 0.68; 95% CI 0.12, 3.83) (Chimbira 1980a).

Weight gain

The trial comparing danazol 200 mg with danazol 100 mg reported the mean gain in weight for the two groups, danazol 200

mg (mean weight gain 2.3 kg after three months of treatment), danazol 100 mg (mean of 2.1 kg) (Chimbira 1980a). No standard deviations were reported in the trial so a WMD could not be calculated.

DANAZOL 200 mg VERSUS REDUCING DOSE DANAZOL

One small parallel trial compared danazol 200 mg with a reducing dose of danazol for three months treatment. Thirty six women were involved in this comparison. The reducing dose danazol was a regime of danazol 200 mg/day for one month, 100 mg/day the next month and danazol 50 mg/day for the third month.

Menstrual blood loss (objective/subjective)

Mean MBL was measured objectively after three months treatment, and there was no significant difference in mean MBL between the two groups (WMD 33.50; 95% CI -32.38, 99.38) (Higham 1993).

Side effects

The number of women in each intervention group experiencing adverse effects was reported as an outcome measure. For this outcome there was no significant difference between the two groups (OR 1.13, 95% CI 0.14, 9.07) (Higham 1993).

Withdrawals due to side effects

The number of withdrawals due to side effects showed no significant difference between the danazol 200 mg group and the group taking a reducing dose of danazol (OR 0.88, 0.15,5.05) (Higham 1993).

Weight gain

The study comparing danazol 200 mg with a reducing dose of danazol assessed the number of women in each group who had a weight gain of >2kg after three months treatment. There was no significant difference between the two groups on this outcome measure (OR 0.32, 95% CI 0.08, 1.28) (Higham 1993).

Duration of menses

The mean duration of menses for women in the two intervention groups was assessed after three months treatment. The only trial in the meta-analysis showed no significant difference between the two intervention groups (WMD 1.30, 95%CI -0.76, 3.36) (Higham 1993).

Subjective efficacy of intervention

Subjective efficacy was measured as the number of women rating the treatment as highly or moderately effective in each of the intervention groups. There was no significant difference between the two groups after three months of daily treatment (OR 1.18, 95% CI 0.30, 4.73) (Higham 1993).

DISCUSSION

The aim of this review was to assess the effectiveness and tolerability of danazol for the treatment of heavy menstrual bleeding. Despite the fact that danazol has been available for the treatment of menorrhagia for a considerable length of time, there is a general lack of well-designed research to evaluate the effectiveness and tolerability of this therapy. The results of this review are based on a small number of trials which are underpowered and with unclear allocation concealment.

The largest trial had 20 women in each arm. A power calculation for sample size based on a (alpha)=0.05 and b (beta)=0.80 where danazol treatment is compared to either progestogen or NSAID therapy (the most common medical treatments) indicated that at least 30 women in each arm would be required to show an acceptable increase in benefit (30%) in the proportion of women having their menstrual bleeding reverting to normal (<80 mls/cycle). It was not feasible to consider a benefit of treatment in terms of the actual quantity of blood loss, mls/cycle, that women find satisfactory and no data were available to assess satisfaction with treatment. Therefore, where no differences between interventions are reported, it may be that the trials are too small to show any difference.

MENSTRUAL BLOOD LOSS: objective measurement

Danazol was shown to be no more effective than progestogens at reducing MBL by one trial included in the meta-analysis. If response to treatment is defined as reduction of menstrual blood loss to <80 ml/cycle, significantly more women in the danazol group had their MBL reduced to below 80 ml/cycle. Another trial, included as descriptive data, suggested that danazol may be more effective than progestogens at reducing MBL. One limitation of the studies comparing danazol with a progestogen is that in all these studies, the progestogen was given in the luteal phase of the cycle (days 19-26). Progestogens given during the luteal phase of the cycle in women with ovulatory HMB may increase, rather than reduce menstrual blood loss (Preston 1995).

Danazol was shown to be more effective at reducing MBL than mefenamic acid by the one trial included in the meta-analysis and two other trials. Danazol was also shown to be more effective than naproxen at reducing MBL by one small trial included in the meta-analysis. Thus there is consistent evidence that danazol is more effective than NSAIDs in reducing MBL.

Danazol was shown to be more effective at reducing MBL than the oral contraceptive pill by one small trial included as Other Data. The results of one trial included as descriptive data indicate danazol is no more effective than a progestogen releasing IUD in reducing MBL.

On the available evidence, a 200 mg daily danazol regime appears to be no more effective in reducing MBL compared to danazol 100 mg or a reducing dose danazol regimen. There are some limitations to this evidence. The results are based on one trial and small numbers of participants for all comparisons and the data for some comparisons are heavily skewed.

It is possible that the differences found were underestimated. Trials were included in this review if women had a subjective complaint of heavy menstrual bleeding and/or if they had objectively determined heavy menstrual bleeding. In all of the included studies, participants had a subjective complaint of heavy menstrual bleeding and in five studies, women had their MBL objectively measured by the alkaline haematin method. To comply with the definition of objectively defined menorrhagia, trials would have to include only women with MBL>80 ml/cycle. However in one study participants were included if their objectively defined MBL was>50 ml/cycle (Cameron 1987) and another included women with a MBL of >60 ml/cycle. Menorrhagia was correctly defined as objectively determined MBL of >80 ml/cycle in three studies (Dockeray 1989; Dunphy 1998; Higham 1993).

Many women who seek medical help for heavy menstrual bleeding have normal blood loss (Fraser 1984; Haynes 1977) and results from one RCT have suggested that there is little response to therapy in women with MBL <35 ml (Fraser 1981). Since a proportion of the study participants with a complaint of heavy menstrual bleeding may have had normal menstrual blood loss, it is likely that some reported differences between treatment and placebo groups have been underestimated.

MENSTRUAL BLOOD LOSS: subjective measurement

Whilst the alkaline haematin extraction method is the most accurate method for assessment of blood loss and is used as the standard, a woman's own perception of her MBL is important in the evaluation of effectiveness of treatment on MBL. Therefore a woman's subjective assessment is an important outcome measure.

Three studies recorded MBL according to subjective measures. One trial compared danazol 200 mg daily and a placebo and the authors reported that danazol was significantly more effective after two months treatment. Danazol was shown to be significantly more effective than progestogens at reducing MBL by one study which used a pictorial bleeding chart method to assess MBL and by one study which used a bleeding intensity score to assess MBL.

The subjective assessment of MBL is an important outcome measure, as most diagnoses and interventions for heavy menstrual bleeding are based on clinical evidence, in the absence of objectively determined heavy menstrual bleeding. It is therefore important in practice that any intervention results in a significant improvement in the woman's own perceived cyclical loss.

MENSTRUAL BLOOD LOSS AFTER CESSATION OF THERAPY

It has been reported that danazol has a significant 'carry-over' effect and many women have reduced MBL for up to four months

after cessation of therapy (Shaw 1994), suggesting that danazol could be used intermittently. Danazol might be more acceptable in clinical use if it could be effectively used intermittently. There are no studies assessing intermittent danazol use for HMB, but two of the included studies (Dunphy 1998; Lamb 1987) assessed whether danazol has a significant 'carry-over' effect. Lamb (1987) reported that a reduction in blood loss score was maintained for four months after treatment ceased, but insufficient information was provided for these data to be included in the review. This finding was not confirmed in one study which compared danazol with a progestogen, and found there to be no significant carry-over effect of danazol 3 months after treatment was stopped (Dunphy 1998).

As danazol has a rapid and significant effect on increasing haemoglobin and serum ferritin (Chimbira 1979; Ford 1994), it may be valuable in women who need a highly effective short term treatment to stop very heavy bleeding and restore their haemoglobin and iron status to normal. One of the potential short term uses of danazol is to use it intermittently, however there is currently insufficient evidence to assess whether danazol can be effectively used on an intermittent basis.

ADVERSE EVENTS

There were no significant differences in reported adverse effects with danazol 200 mg in comparison with a reducing dose danazol. No data are available for whether women treated with danazol experience more adverse effects than those treated with the oral contraceptive or a progestogen releasing IUD. However, women treated with danazol experienced significantly more adverse effects than those treated with a progestogen and those treated with mefenamic acid.

Danazol has weak androgenic properties, and hence can cause side effects related to this. These side effects spontaneously resolve after cessation of treatment. None of the included trials reported data on numbers experiencing specific adverse events in a suitable form for inclusion in this review, but commonly reported side effects for danazol treatment included acne, weight gain, headache, nausea and tiredness. One study comparing danazol with mefenamic acid, reported that women in the mefenamic acid group experienced mostly gastrointestinal side effects such as nausea/vomiting and diarrhoea, whereas women in the danazol group experienced more serious side effects such as musculoskeletal pains, flushes, behavioural changes, lethargy, and androgenic effects such as acne, breast atrophy, hirsutism and hoarseness (Dockeray 1989).

Women using danazol for heavy menstrual bleeding are likely to require long term treatment, so any adverse events which affect adherence to treatment or treatment safety are particularly important. One study reported that for the women treated with danazol, the side effects of breast atrophy, hirsutism and hoarseness did not develop until the second month of treatment (Dockeray 1989). As the studies included in this review all involved two to

three months treatment, data on the side effect profile of danazol for longer term treatment with danazol would be useful. A longer period of treatment would enable the longer term liver effects of danazol to be assessed. These potential liver effects may be an important factor limiting the long term use of danazol and they have not been evaluated by any of the studies.

WEIGHT GAIN

Weight gain is one of the androgenic side effects of danazol which may limit its use (Irvine 1999), and is therefore an important outcome measure. When compared to placebo, the mean weight of the danazol group was significantly greater than that of the placebo group after two months treatment. Three studies assessed weight gain for danazol compared with a progestogen. The mean weight gain of the danazol group was significantly greater than that of the progestogen group. However when weight gain was assessed in terms of the number of women with a weight gain of > 2kg (Higham 1993) and the number of women with a weight gain of >3kg (Bonduelle 1991), there was no significant difference between the two interventions. There was no significant difference in terms of the number of women with weight gain of >2kg when danazol 200mg was compared with a reducing dose of danazol. However assessment of weight gain by a dichotomous outcome may not be sensitive to determine real differences between the groups. One study compared mean weight gain for danazol 200mg versus 100 mg. There appears to be no difference in terms of mean weight gain between these regimens, however there is currently insufficient evidence to answer this question.

WITHDRAWALS DUE TO SIDE EFFECTS

There was no significant difference in withdrawals due to side effects when danazol 200 mg daily was compared to placebo, progestogens or a reducing dose danazol. However, due to the small number of women involved in the studies, there is insufficient evidence to evaluate this outcome adequately. It may be that the numbers are too small to reveal any differences between the groups. The large drop out rates of several trials (five studies had drop out rates of more than 10%) points towards the treatment being unacceptable to women. Some of these may be unreported withdrawals due to side effects.

DYSMENORRHOEA

There was no significant difference between danazol 200 mg/day, NSAIDs and a reducing dose of danazol in terms of the number of women reporting an improvement in dysmenorrhoea. There is insufficient evidence to determine whether there is any difference between danazol and progestogens with regards to improvement in dysmenorrhoea. One study has evaluated this, and the results indicate that progestogens may produce a greater improvement in dysmenorrhoea (Bonduelle 1991).

DURATION OF MENSES

A significant reduction in the duration of menses in favour of danazol is shown for those trials comparing danazol with NSAIDs and a progesterone releasing IUD. The results of one trial indicate that the duration of menses may be shorter after treatment with a 200mg danazol regimen when compared with a 100 mg regimen. Another study showed that the duration of menses is significantly shorter for women treated with a reducing dose of danazol compared to those treated with danazol 200 mg/day. There is no significant difference in the duration of menses when danazol is compared to progestogens. The results from one trial comparing danazol with a progestogen, indicate that danazol may cause substantially more protracted bleeding in some women, whereas this is not the case for norethisterone (Higham 1993). The reason for this is unknown.

EFFICACY OF INTERVENTION

Two studies comparing danazol with a progestogen evaluated subjective efficacy of treatment. One study showed subjective efficacy was significantly greater for danazol, where as the second found no difference between the two interventions.

ACCEPTABILITY OF INTERVENTION

One study comparing danazol and mefenamic acid assessed the acceptability of treatment, and found there was no difference between the two treatments in terms of the number of women unwilling to continue treatment. However, it may be that too few participants were included in the analysis to show any difference between the groups. The study reported that the reason women were unwilling to continue treatment differed between the two groups. For the mefenamic acid group, the reason most women gave for being unwilling to continue treatment was due to a lack of efficacy, whereas for women in the danazol group it was largely due to side effects (Dockeray 1989).

DOSAGE

One of the objectives of this review is to determine the optimum dosage of danazol as a treatment for HMB. Two small trials compared the standard dose of danazol for HMB, 200 mg/day with a lower dose of 100 mg and a reducing dose regimen. No differences in effectiveness or frequency of adverse events were reported, but women treated with 200 mg/day had a shorter duration of menses when compared with a reducing dose regimen. Numbers of women in the trials were insufficient to adequately assess this outcome.

It is important to note that there are no studies comparing danazol with tranexamic acid and the levonorgestrel-releasing intrauterine system (Mirena). These treatments may be as effective as danazol, but an objective comparison has not been carried out.

No study has included changes in quality of life or resource cost as outcome measures. Danazol has a contraceptive effect in doses of above 200 mg/day, however it is not licensed for use as a contracep-

tive, and therefore women not wishing to conceive require additional contraception. This is particularly important with danazol as it is has a teratogenic effect and there is a risk of masculinisation of a female fetus if exposure to danazol is continued between 8 and 18 weeks of gestation. This means that the acceptability of danazol to women, and quality of life outcomes are very important and they have not been properly addressed by any of the trials.

SUMMARY OF THESE RESULTS IN TERMS OF THE OBJECTIVES

1. Is danazol more effective than placebo in reducing heavy menstrual blood loss?

One small trial assessed this outcome, and danazol appears to be more effective than placebo, but the data are poorly reported.

2. Is danazol more effective than other medical therapies?

Danazol is more effective than progestogens, NSAIDs and the OCP, although the results are imprecise with wide confidence intervals

3. What is the optimum dosage of danazol?

The standard dose of 200 mg/day of danazol does not appear to differ in effectiveness and frequency of adverse events when compared to 100 mg/day or a reducing regimen. However these results are based on one small trial for each comparison.

4. Does treatment with danazol lead to an improved quality of life for women with HMB?

The included trials only assessed improvement in dysmenorrhoea and there were insufficient data to address this outcome adequately.

5. Do women tolerate treatment with danazol and find it acceptable?

There was an increased frequency of adverse events when danazol was compared with other medical therapies. This did not appear to affect the acceptability of the treatment. However this comparison was only reported by one small study.

AUTHORS' CONCLUSIONS

Implications for practice

The results do not give clear indications for recommending danazol as a treatment for heavy menstrual bleeding. Danazol appears to be an effective treatment for heavy menstrual bleeding when compared to other medical treatments, though it is uncertain whether it is acceptable to women. The use of danazol may be limited by its side effect profile, its acceptability to women and the need for continuing treatment.

Implications for research

Additional well designed RCTs with sufficient power are needed to test the efficacy of danazol compared to other medical therapies. Future trial design needs to include outcomes such as quality of life measures and a longer period of treatment (at least six months) to adequately evaluate adverse events and participant satisfaction. However, there may be difficulties in doing long term research because of the side effects of danazol and the existence of more acceptable alternatives.

ACKNOWLEDGEMENTS

The authors acknowledge the helpful comments of those who refereed previous versions of this review. Special thanks are due to Michelle Proctor, Review Group Coordinator, for her help with all the inevitable problems; to Ruth Withers, Trials Search Coordinator, for her assistance with identifying trials; and to Sue Hall, Secretary of the Review Group, for her secretarial help. The authors would also like to acknowledge the input of Metin Gulmezoglu for his help in translating the study published in Turkish.

REFERENCES

References to studies included in this review

Bonduelle 1991 {published data only}

Bonduelle M, Walker JJ, Calder AA. A comparative study of Danazol and Norethisterone in dysfunctional uterine bleeding presenting as menorrhagia. *Postgraduate Medical Journal* 1991;67:833–836.

Buyru 1995 {published data only}

Buyru F, Yalcin O, Kovanci E, Turfanda A. Danazol treatmant for dysfunctional uterine bleeding [Disfonksiyonel uterin kanamalarda danazol tedavisi]. *Instanbul Tip fakultesi Mecmuasi* 1995;**58**(3):37–40.

Cameron 1987 {published data only}

Cameron IT. Dysfunctional uterine bleeding. *Bailliere's Clinical Obstetrics and Gynaecology* 1989;**3**(2):315–327.

* Cameron IT, Leask R, Kelly RW, Baird DT. The effects of danazol, mefenamic acid, noerethisterone and a progesterone-impregnated coil on endometrial prostaglandin concentrations in women with menorrhagia. *Prostaglandins* 1987;**34**(1):99–110.

Chimbira 1980a {published data only}

Chimbira TH, Anderson ABM, Naish C, Cope E, Turnbull AC. Reduction of menstrual blood loss by Danazol in unexplained menorrhagia: Lack of effect of placebo. *British Journal of Obstetrics and Gynaecology* 1980;**87**:1152–1158.

Dockeray 1989 {published data only}

Dockeray CJ, Sheppard BL, Bonnar J. Comparison between mefenamic acid and danazol in the treatment of established menorrhagia. *British Journal of Obstetrics and Gynaecology* 1989;**96**:840–844.

Dunphy 1998 {published data only}

Dunphy BC, Goerzen J, Greene CA, De La Ronde S, Seidel J, Ingelson B. A double blind randomised study comparing danazol and medroxyprogesterone acetate in the management of menorrhagia. *Journal of Obstetrics and Gynaecology* 1998;**18**(6):553–555.

Fraser 1991 {published and unpublished data}

Fraser IS, McCarron G. Randomized trial of 2 hormonal and prostaglandin-inhibiting agents in women with a complaint of menorrhagia. *Australian and New Zealand Journal of Obstetrics & Gynaecology* 1991;**31**(1):66–70.

Higham 1993 {published and unpublished data}

Higham JM, Shaw RW. A comparative study of Danazol, a regimen of decreasing doses of danazol, and norethindrone in the treatmant of objectively proven unexplained menorrhagia. *American Journal of Obstetrics & Gynecology* 1993;**169**:1134–9.

Lamb 1987 {published data only}

Lamb MP. Danazol in menorrhagia: a double blind placebo controlled trial. *Journal of Obstetrics and Gynaecology* 1987; 7:212–216.

References to studies excluded from this review

Chimbira 1980b

Chimbira TH, Anderson ABM, Naish C, Cope E, Turnbull AC. Reduction of menstrual blood loss by Danazol in unexplained menorrhagia: Lack of effect of placebo. *British Journal of Obstetrics and Gynaecology* 1980;**87**:1152–1158.

Need 1992

Need JA, Forbes KL, Milazzo L, McKenzie E. Danazol in the treatment of menorrhagia: the effect of a 1 month induction dose (200mg) and 2 months maintenance therapy (200mg, 100mg or placebo). *Australian and New Zealand Journal of Obstetrics & Gynaecology* 1992;**32**(4):346–352.

Additional references

Altman 1996

Altman DG, Bland JM. Detecting skewness from summary information. *British Medical Journal* 1996;**313**:1200–1200.

Bradlow 1992

Bradlow J, Coulter A, Brooks P. *Patterns of referral*. Oxford: Oxford Health Services Research Unit, 1992.

Chimbira 1979

Chimbira TH, Cope E, Anderson AB, Bolton FG. The effect of danazol on menorrhagia, coagulation mechanisms, haematological indices and body weight. *British Journal of Obstetrics & Gynaecology* 1979;**86**(1):46–50.

Chimbira 1980

Chimbira TH, Anderson AB, Cope E, Turnbull AC. Effect of danazol on serum gonadotrophins and steroid hormone concentrations in women with menorrhagia. *British Journal of Obstetrics & Gynaecology* 1980;**87**(4):330–336.

Chimbira 1980c

Chimbira TH, Anderson ABM, Turnbull AC. Relation between measured menstrual blood loss and patient's subjective assessment of loss, duration of bleeding, number of sanitary towels, uterine weight and endometrial surface area. *British Journal of Obstetrics and Gynaecology* 1980;87: 603–609.

Clarke 1995

Clarke A, Black N, Rowe P, Mott S, Howle K. Indications for and outcome of total abdominal hysterectomy for benign disease: a prospective cohort study. *British Journal of Obstetrics & Gynaecology* 1995;**102**:611–620.

Cole 1971

Cole SK, Billewize WZ, Thomson AM. Sources of variation in menstrual blood loss. *Journal of Obstetrics & Gynaecology of British Commonwealth* 1971;**78**:933–939.

Coulter 1991

Coulter A, Bradlow J, Agass M, Martin-Bates C, Tulloch A. Outcomes of referrals to gynaecology out-patients clinics for menstrual problems: an audit of general practice records. *British Journal of Obstetrics & Gynaecology* 1991; **98**:789–796.

Coulter 1994

Coulter A, Peto V, Jenkinson C. Quality of life and patient satisfaction following treatment for menorrhagia. *Family Practice* 1994;**11**:394–401.

Coulter 1995

Coulter A, Kelland J, Peto V, Rees MC. Treating menorrhagia in primary care. An overview of drug trials and a survey of prescribing practice. *International Journal of Technology Assessment in Health Care* 1995;**11**:456–471.

Dicker 1982

Dicker RC, Greenspan JR, Strauss LT, Cowart MR, Scally MJ, Peterson HB, et al. Complications of abdominal and vaginal hysterectomy among women of reproductive age in the Unites States. The Collaborative Review of Sterilization. *American Journal of Obstetrics & Gynaecology* 1982;**144**: 841–848.

EHCB 1995

Anonymous. The management of menorrhagia. Effective Health Care Bulletin 1995; Vol. 9.

Ford 1994

Ford I, Li TC, Cooke ID, Preston FE. Changes in haematological indices, blood viscosity and inhibitors of coagulation during treatment of endometriosis with danazol. *Thrombosis & Haemostasis* 1991;**72**(2):218–221.

Fraser 1981

Fraser IS, Pearse C, Shearman RP, Elliott PM, McIlveen J, Markham R. Efficacy of mefenamic acid in patients with a complaint of menorrhagia. *Obstetrics & Gynaecology* 1981; **58**:543–551.

Fraser 1984

Fraser IS, McCarron G, Markham R. A preliminary study of factors influencing perception of menstrual blood loss volume. *American Journal of Obstetrics and Gynaecology* 1984;**149**:788–793.

Gath 1982

Gath D, Cooper P, Day A. Hysterectomy and psychiatric disorder. I: Levels of psychiatric morbidity before and after hysterectomy. *British Journal of Psychiatry* 1982;**140**: 335–342.

Hallberg 1964

Hallberg L, Nilsson L. Determination of menstrual blood loss. *Scandanavian Journal of Laboratory Investigation* 1964; **16**:244–248.

Hallberg 1966

Hallberg L, Hogdahl A, Nilsson L, Rybo G. Menstrual blood loss - a population study: variation at different ages and attempts to define normality. *Acta Obstetrica et Gynecologica Scandinavica* 1966;**45**:320–351.

Haynes 1977

Haynes PJ, Hodgson H, Anderson AB, Turnbull AC. Measurement of menstrual blood loss in patients complaining of menorrhagia. *British Journal of Obstetrics & Gynaecology* 1977;**84**(10):763–8.

Higham 1990

Higham JM, O'Brien PMS, Shaw RW. Assessment of menstrual blood loss using a pictorial chart. *British Journal of Obstetrics and Gynaecology* 1990;**97**:734–739.

Hospital 1995

HMSO. Finished consultant episodes by diagnosis, operation and speciality. Hospital Episode Statistics 1995; Vol. Volume 1.

Irvine 1999

Irvine GA, Cameron IT. Medical management of dysfunctional uterine bleeding. *Balliere's Clinical Obstetrics and Gynaecology* 1999;**13**(2):189–202.

Preston 1995

Preston JT, Cameron IT, Adams EJ, Smith SK. Comparative study of tranexamic acid and norethisterone in the treatment of ovulatory menorrhagia. *British Journal of Obstetrics and Gynaecology* 1995;**102**:410–416.

Shaw 1994

Shaw RW. Assessment of medical treatments for menorrhagia. *British Journal of Obstetrics and Gynaecology* 1994;**101**(supp 11):15–18.

Vessey 1992

Vessey MP, Villard-Mackintosh L, McPherson K, Coulter A, Yeates D. The epidemiology of hysterectomy: findings in a large cohort study. *British Journal of Obstetrics & Gynaecology* 1992;**99**:402–407.

* Indicates the major publication for the study

SOURCES OF SUPPORT

External sources of support

• No sources of support supplied

Internal sources of support

- Department of Obstetrics and Gynaecology, University of Auckland NEW ZEALAND
- Royal College of Obstetricians and Gynaecologists UK

INDEX TERMS

Medical Subject Headings (MeSH)

Danazol [*therapeutic use]; Estrogen Antagonists [*therapeutic use]; Menorrhagia [*drug therapy]; Randomized Controlled Trials

MeSH check words

Female; Humans